Author's response to reviews

Title: L1CAM is preferentially expressed in triple-negative breast cancers and is inversely correlated with Androgen receptor

Authors:

Kai Doberstein KD (k.doberstein@dkfz.de)
Karin Milde-Langosch KML (milde@uke.de)
Niko P Bretz NPB (NBRETZ@partners.org)
Uwe Schirmer US (u.schirmer@dkfz.de)
Ayelet Harari AH (Ayelethar@clalit.org.il)
Isabell Witzel IW (jwitzel@uke.de)
Alon Ben-Arie ABA (abenarie@hotmail.com)
Michael Hubalak MH (Michael.hubalek@i-med.ac.at)
Elisabeth Müller-Holzner EMH (Elisabeth.mueller@i-med.ac.at)
Alain G Zeimet AGZ (Alain.zeimet@i-med.ac.at)
Peter Altevogt PA (p.altevogt@dkfz.de)
Mina Fogel MF (Mina_F@clalit.org.il)

Version: 2
Date: 8 September 2014

Author's response to reviews:

not applicable. Transfer from BCR.